Author's response to reviews

Title: A three-dimensional model of error and safety in surgical health care microsystems. Rationale, development and initial testing.

Authors:

Peter McCulloch (peter.mcculloch@nds.ox.ac.uk)
Ken Catchpole (ken.catchpole@nds.ox.ac.uk)

Version: 5 Date: 17 August 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team

In response to your e-mail of 12th August we have made the following changes, which you will find in the uploaded revision. The text in CAPS is our response to the sections of your e-mail (lower case). I hope this is clear, but please contact me if not.

Major revisions (we require the author to make these changes)

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions

MODIFIED AS REQUESTED

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract, Background, Results, Discussion, Conclusions, Methods, Authors' contributions, Authors¿ information (optional), Acknowledgements (optional), References

IN FACT THIS HAD ALREADY BEEN DONE

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

CONFIRMATION OF THIS FACT ADDED

References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].`

DONE AS REQUESTED

References: please provide a functioning link to website http://csru.org.uk

LINK UPDATED AND CORRECTED AS REQUESTED
References: Please remove your references and figure legend from the box and place as plain text

WE DO NOT UNDERSTAND THIS COMMENT. BOTH SECTIONS ARE ALREADY IN PLAIN TEXT

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.

DONE AS REQUESTED

Tables- please remove the visible vertical lines from your tables

DONE AS REQUESTED

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

DONE AS REQUESTED

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

DONE AS REQUESTED

Yours Sincerely

Peter McCulloch